Author's response to reviews

Title: Circulating Angiopoietins-1 and -2, Angiopoietin Receptor Tie-2 and Vascular Endothelial Growth Factor-A as Biomarkers of Acute Myocardial Infarction: a Prospective Nested Case-Control Study

Authors:

Carlos - Iribarren (cgi@dor.kaiser.org)
Bruce H Phelps (bruce.phelps@aviir.com)
Jeanne A Darbinian (jad@dor.kaiser.org)
Edward R McCluskey (ted.mccluskey@aviir.com)
Charles P Quesenberry (cpq@dor.kaiser.org)
Evangelos Hytopoulos (vangelis.hytopoulos@aviir.com)
Joseph H Vogelman (dr.jhv@juno.com)
Norman Orentreich (dr.jhv@juno.com)

Version: 3 Date: 10 June 2011

Author's response to reviews:

Author list formatted as requested
Affiliations as requested
Author contributions formatted as requested
Abbreviations at end
Competing interests formatted as requested
Deleted vertical lines from tables